MATHEMATICAL TRIPOS Part III

Monday, 8 June, 2009 1:30 pm to 3:30 pm

PAPER 59

INTRODUCTION TO TWISTOR THEORY

This is a two hour paper with FOUR questions

of whichTHREE are to be attempted.

Each question should take about 40 minutes. Each question is marked out of 33 possible points.

There is one mark for a particularly good bit of proof or comment available across al three questions This gives a total of 100 possible marks.

STATIONERY REQUIREMENTS SPECIAL REQUIREMENTS Cover sheet None Treasury Tag Script paper

You may not start to read the questions printed on the subsequent pages until instructed to do so by the Invigilator.

Part III, Paper 59

Part III, Paper 59 [TURN OVER

END OF PAPER

Part III, Paper 59